Author's response to reviews

Title: Systematic Development of the YouRAction program, a computer-tailored Physical Activity promotion intervention for Dutch adolescents, targeting personal motivations and environmental opportunities.

Authors:

Richard G Prins (r.prins@erasmusmc.nl)
Pepijn van Empelen (p.vanempelen@erasmusmc.nl)
Mariëlle A Beenackers (m.beenackers@erasmusmc.nl)
Johannes Brug (j.brug@vumc.nl)
Anke Oenema (a.oenema@erasmusmc.nl)

Version: 2 Date: 5 August 2010

Author's response to reviews:

BMC Public Health

August 5, 2010

Dear dr. Melissa Norton,

Thank you for accepting our paper and for your comments to format the manuscript. For the latest version of the manuscript following changes have been made:

Your comment: Authors' contributions: Please indicate that all authors read and approved the final manuscript.

Our response: we have added the suggested sentence

Your comment: Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system

Our response: we have added the suggested details of additional files and removed additional files which were not mentioned in text from the system

Your comment: Word counts: please remove these.

Our response: we have removed word counts

Your comment: Line numbers: please delete these.

Our response: we have removed line numbers

Your comment: Placement text: please delete all figure and table placement text.
Our response: we have removed the placement texts

Your comment: Tables- please remove the visible vertical lines from your tables
Our response: we have removed the vertical lines

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows : please re-layout your table without these merged elements
Our response: we have re-layouted the tables

Your comment: Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Our response: we have checked the manuscript and made some typographical changes.

Sincerely yours on behalf of all authors,

R.G. Prins, MSc
Erasmus Medical Center
Department of Public Health
PO box 2040
3000 CA Rotterdam
r.prins@erasusmc.nl
phone: +31 10-70 43721
fax: +31 10-70 38474.